\documentclass[12pt]{article}
\usepackage{fullpage,enumitem,amsmath,amssymb,graphicx}
\usepackage{graphicx} % This is a package for including graphics in your solution.


\begin{document}

\begin{center}
{\Large CS265/CME309, Problem Set [number]}

\begin{tabular}{rl}
SUNet ID(s): & \\
Name(s): & \\
\end{tabular}
\end{center}

By turning in this assignment, I agree by the Stanford honor code and declare that all of this is the work of my partner and I.

\section*{Problem 1}

\begin{enumerate}[label=(\alph*)]
  \item (your solution)
  \item (your solution)
\end{enumerate}

\section*{Problem 2}

\begin{enumerate}[label=(\alph*)]
  \item (your solution, here is a high-level algorithm description in fancy font)
\begin{verbatim}
Algorithm for Pset:
    solve problem
    output ``answer''
\end{verbatim}

  \item (your solution)
\end{enumerate}

\end{document}